Author's response to reviews

Title: Implementation of tuberculosis infection control in health facilities in Mukono and Wakiso districts, Uganda

Authors:

   Esther EB Buregyeya (eburegyeya@musph.ac.ug)
   Suzanne SV Verver (VerverS@kncvtbc.nl)
   Bart BC Criel (BCriel@itg.be)
   Fred FN Nuwaha (fnuwaha@musph.ac.ug)
   Robert RC Colebunders (bcoleb@itg.be)
   Rhoda RW Wanyenze (rwanyenze@musphcdc.ac.ug)
   Joan N Kalyango (nakayaga2001@yahoo.com)
   Achilles AK Katamba (akatamba@yahoo.com)
   Ellen MH Mitchell (MitchellE@kncvtbc.nl)

Version: 2 Date: 5 October 2012

Author's response to reviews:

5/10/2012
The Chief Editor
BMC Infectious Diseases Journal,
Dear Sir/Madam,

RE: Implementation of tuberculosis infection control in health facilities in Mukono and Wakiso districts, Uganda (Manuscript ID 1729142150819541)

I am sorry, I submitted the manuscript with a wrong referencing system. I have not corrected it and am submitting a revised version.

Thank you.

Yours sincerely,

Esther Buregyeya
Lecturer,
Department of Disease Control & Environmental Health
Makerere University School of Public Health
P.O.Box 7062, Kampala, Uganda.
E-mail: eburegyeya@musph.ac.ug